Author's response to reviews

Title: Dissemination and implementation of suicide prevention training in one Scottish region

Authors:

Linda Gask (Linda.Gask@manchester.ac.uk)
Gillian Lever-Green (g.d.lever-green@manchester.ac.uk)
Rebecca Hays (Rebecca.Hays@manchester.ac.uk)

Version: 5 Date: 1 December 2008

Author's response to reviews:

we have removed the early reference to table 4, which will solve the problem highlighted by the editors of the citation of table 4 before tables 2 and 3.